Author’s response to reviews

Title: The minor C-allele of rs2014355 in ACADS is associated with reduced insulin release after an oral glucose load

Authors:

Malene Hornbak (mnhr@hagedorn.dk)
Karina Banasik (kabs@hagedorn.dk)
Johanne M Justesen (jmju@hagedorn.dk)
Nikolaj T Krarup (ntk@hagedorn.dk)
Camilla H Sandholt (cila@novonordisk.com)
Åsa Andersson (aaa@farma.ku.dk)
Anelli Sandbæk (annelli.sandbaek@alm.au.dk)
Torsten Lauritzen (tl@alm.au.dk)
Charlotte Pisinger (chpi@glo.regionh.dk)
Daniel R Witte (drw@steno.dk)
Thorkild I.A. Sørensen (tias@ipm.regionh.dk)
Oluf Pedersen (oluf@hagedorn.dk)
Torben Hansen (toha@hagedorn.dk)

Version: 3 Date: 14 December 2010

Author’s response to reviews: see over
Dear Editor,

Thank you for accepting our manuscript for publication in BMC Medical Genetics. In the resubmitted manuscript we have addressed the following comment:

Ethics
Please name the specific ethics committee which approved your research and include the name in your manuscript.
This has been included in the manuscript on page 6, line 142-143.

Sincerely yours,

MSc. Karina Banasik
Hagedorn Research Institute
Niels Steensens Vej 1, NLD2.22
DK-2820 Gentofte, Denmark.